
diff -r ef5f62bf5987 -r 9a04eb6a2291 cws/cw01.tex
--- a/cws/cw01.tex	Tue Oct 29 23:56:13 2019 +0000
+++ b/cws/cw01.tex	Wed Oct 30 11:28:44 2019 +0000
@@ -14,12 +14,11 @@
 
  
 \noindent
-This assignment is about Scala and worth 10\%. The preliminary
-part is due on \cwSIX{} at 4pm, and the core part on \cwSIXa{}
-at 4pm. You are asked to implement two programs about list
-processing and recursion. The core part is more advanced and might
-include material you have not yet seen in the first lecture.
-\bigskip
+This assignment is about Scala. You are asked to implement two programs
+about list processing and recursion. The preliminary part (3\%) is due
+on \cwSIX{} at 4pm, and the core part on \cwSIXa{} at 4pm.  The core
+part is more advanced and might include material you have not yet seen
+in the first lecture.\bigskip
  
 \IMPORTANT{}
 




